Reviewer #1:
Scientific Quality: Grade C (Good)
Language Quality: Grade B (Minor language polishing)
Conclusion: Minor revision
Specific Comments to Authors: This paper is well written and interesting. However, I suggest that enteral nutrition should be mentioned and discussed in the paper.

Reply: Thank you for your comments. We added informations about enteral nutrition.

Reviewer #2:
Scientific Quality: Grade C (Good)
Language Quality: Grade B (Minor language polishing)
Conclusion: Accept (General priority)
Specific Comments to Authors: Minor language polishing would be necessary.

Reply: Thank you for your comments. We has revised the language as suggested.